Author's response to reviews

Title: Differential regulation of MMPs by E2F1, Sp1 and NF-kappaB controls the small cell lung cancer invasive phenotype

Authors:

Zunling Li (lizunling@bzmc.edu.cn)
Yanxia Guo (qingfeng19870605@126.com)
Hanming Jiang (hmjiang@tsmc.edu.cn)
Tingguuo Zhang (tingguo_zhang@yahoo.cn)
Changzhu Jin (jincz63@hotmail.com)
Charles Y F Young (youngc@mayo.edu)
Huiqing Yuan (lyuanhq@sdu.edu.cn)

Version: 4 Date: 4 April 2014

Author's response to reviews: see over
Dear Ms. Cherry Battad,

Good day!

We are very pleased when we read your E-mail. The funding source for each individual author has been added in the acknowledgement section in our revised manuscript.

Thank you for your work.

Best regards.

Huiqing Yuan